Author's response to reviews

Title: Does Glycine max leaves or Garcinia Cambogia promote weight-loss or lower plasma cholesterol in overweight individuals: A randomized control trial

Authors:

Ji-Eun Kim (jek8604@hanmail.net)
Seon-Min Jeon (smjeon@knu.ac.kr)
Ki Hun Park (khpark@gsmu.ac.kr)
Woo Song Lee (wslee@kribb.re.kr)
Tae-Sook Jeong (tsjeong@kribb.re.kr)
Robin A McGregor (robinmcgregor@gmail.com)
Myung-Sook Choi (mschoi@knu.ac.uk)

Version: 3 Date: 20 September 2011

Author's response to reviews:

September 20th, 2011

Myung-Sook Choi, Ph.D.
Department of Food and Nutrition
Kyungpook National University
Daegu 702-701, Korea
Tel: +82-53-950-6232
Fax: +82-53-958-1230
E-mail: mschoi@knu.ac.kr

Dear Nehme Gabriel (Editor-in-Chief)

We have made the following formatting changes as requested:

Major revisions (we require the author to make these changes)

References: All references should be cited in the text and should be cited using consecutive numbers in square brackets. Currently, reference 16 is the next cited after reference 14. Also, please place the reference before the full stop. For example `text [1].'

1) We have now corrected references, it appears reference 15 was inadvertently deleted last time.
2) Table 4 and Table 5 appeared to have been inadvertently duplicated during editing of the revised manuscript, we have replace Table 5 with the correct data from the original submitted manuscript.

Minor revisions (we can make these changes for you, although it will speed up
publication of your manuscript if you do them while making the major changes above)

1) Line numbers were removed.

2) All forms of tracking and coloured text were removed.

3) Typographical errors are now corrected.

4) Table titles are now all 12 point.

5) Missing abbreviations were added below tables.

6) We changed the page orientation of Table 6, to be consistent with all the other tables.

Kind Regards
Myung-Sook Choi, Ph.D.